**Author's response to reviews**

**Title:** The association between RCAS1 expression in laryngeal and pharyngeal cancer and its healthy stroma with cancer relapse

**Authors:**

- Magdalena Dutsch-Wicherek (mowicher@gmail.com)
- Romana Tomaszewska (romatom@mp.pl)
- Agata Lazar (agalaz@interia.pl)
- Jacek Skladzien (mowicher2@autograf.pl)
- Lukasz Wicherek (mowicher@cyf-kr.edu.pl)

**Version:** 2 **Date:** 11 December 2008

**Author's response to reviews:** see over
Dec 12, 2008

BMC Cancer

Editor-in-Chief

Dear Editor-in-Chief

I would like to resubmit a revised version of the manuscript entitled: “The association between RCAS1 expression in laryngeal and pharyngeal cancer and its healthy stroma with cancer relapse” to the BMC Cancer, which we have revised in accordance with the Reviewers helpful suggestions. We are pleased, that all suggestions and requests for clarification could be incorporated into a final manuscript.

With best regards,

Magdalena Dutsch-Wicherek, M.D., Ph.D

Otolaryngology Department

Jagiellonian University

Krakow, Poland

mowicher@cyf-kr.edu.pl